
diff -r 38e368dc9b10 -r a6a6bf32fade handouts/ho08.tex
--- a/handouts/ho08.tex	Wed Nov 18 03:02:28 2015 +0000
+++ b/handouts/ho08.tex	Fri Nov 20 02:23:24 2015 +0000
@@ -132,9 +132,9 @@
 a \pcode{pop}-instruction is needed. 
 
 \begin{lstlisting}[language=JVMIS, numbers=none,mathescape]
-$\textrm{\textit{estimate}}($exp1$)$
+$\textrm{\textit{compile}}($exp1$)$
 pop
-$\textrm{\textit{estimate}}($exp2$)$
+$\textrm{\textit{compile}}($exp2$)$
 \end{lstlisting}
 
 \noindent In effect we ``forget'' about the result the first
@@ -169,7 +169,7 @@
 
 
 \begin{lstlisting}[language=JVMIS, numbers=none,mathescape]
-$\textrm{\textit{estimate}}($1+2$)$
+$\textrm{\textit{compile}}($1+2$)$
 dup
 invokestatic XXX/XXX/write(I)V
 \end{lstlisting}




